Title: Trial of Improved Practices approach to explore the acceptability and feasibility of different modes of Chlorhexidine application for neonatal cord care in Pemba, Tanzania

Authors:

- Usha Dhingra (udhingra@jhu.edu)
- Sunil Sazawal (ssazawal@jhu.edu)
- Pratibha Dhingra (pdhingra@cphealthkinetics.in)
- Arup Dutta (adutta@cphealthkinetics.org)
- Said Muhammed Ali (saidmali2003@yahoo.com)
- Shaali Makame Ame (shaaliame@yahoo.com)
- Saikat Deb (saikatdeb@gmail.com)
- Atifa Moh'd Suleiman (atifasule@yahoo.com)
- Robert E Black (rblack1@jhu.edu)

Version: 2  Date: 9 October 2015

Author's response to reviews:

Separate covering letter is attached